

Melatonin versus midazolam in the premedication of anxious children attending for elective surgery under general anaesthesia: the MAGIC non-inferiority RCT

Christopher Deery⁰,^{1*} Robert Bolt⁰,¹ Diana Papaioannou⁰,² Matthew Wilson⁰,³ Marie Hyslop[®],² Esther Herbert[®],² Nikki Totton[®],² Zoe Marshman[®],¹ Tracey Young[®],³ Jennifer Kettle[®],¹ Sondos Albadri[®],⁴ Simon Atkins[®],¹ Katie Biggs[®],² Janet Clarkson[®],⁵ Chris Evans[®],⁶ Laura Flight[®],³ Jacqui Gath[®],⁷ Fiona Gilchrist[®],¹ Kate Hutchence[®],² Nicholas Ireland[®],⁸ Amanda Loban[®],² Amy Norrington[®],⁹ Hamish Paton,¹⁰ Jaydip Ray[®],¹¹ Helen Rodd^{0,1} Elena Sheldon^{0,2} Richard Simmonds⁰² and Christopher Vernazza¹²

²Sheffield Clinical Trials Research Unit, Sheffield Centre for Health & Related Research, University of Sheffield, Sheffield, UK

- ³Sheffield Centre for Health & Related Research, University of Sheffield, Sheffield, UK
- ⁴School of Dentistry, University of Liverpool, Liverpool, UK
- ⁵Dundee Dental Hospital and School, University of Dundee, Dundee, UK
- ⁶University College London Hospitals NHS Foundation Trust, London, UK
- ⁷Patient & Public Involvement, London, UK
- ⁸Newcastle upon Tyne Hospitals NHS Foundation Trust, Newcastle upon Tyne, UK

⁹South Tees Hospitals NHS Foundation Trust, Middlesbrough, UK

- ¹⁰Barnsley Hospital NHS Foundation Trust, Barnsley, UK
- ¹¹Division of Neuroscience, University of Sheffield, Sheffield, UK
- ¹²School of Dental Sciences, Newcastle University, Newcastle upon Tyne, UK

*Corresponding author c.deery@sheffield.ac.uk

Published July 2025 DOI: 10.3310/CWKF1987 Volume 29 • Issue 29

Abstract

Background: Anxiety in children prior to general anaesthesia is common, with up to half displaying distress. Anxiety and distress may lead to unsuccessful anaesthesia, together with greater postoperative pain, agitation and behavioural changes after surgery including sleep disturbances. Midazolam is the current standard premedication; however, it has adverse effects such as the potential for respiratory suppression and unpredictable effects which may result in agitation rather than anxiolysis. Melatonin is an alternative preoperative anxiolytic; however, previous trials have delivered conflicting results. The aim of this non-inferiority trial was to evaluate the effectiveness of melatonin compared to midazolam in reducing anxiety in children undergoing general anaesthesia.

Methods: We undertook a randomised-controlled, parallel-group, double-blind, non-inferiority trial in 20 United Kingdom National Health Service trusts, with an embedded qualitative study and health economic evaluation. Anxious children having day case elective surgery under general anaesthesia were randomly assigned to either control (standard of care) group: midazolam; or intervention group: melatonin. The primary outcome was preoperative distress (non-inferiority hypothesis) as assessed by modified Yale Preoperative Anxiety Scale Short Form. Secondary outcomes included safety and efficacy objectives. Analyses were by intention to treat, with an additional per-protocol analysis. The sample size of the trial was 624 children.

¹School of Clinical Dentistry, University of Sheffield, Sheffield, UK

This synopsis should be referenced as follows: Deery C, Bolt R, Papaioannou D, Wilson M, Hyslop M, Herbert E, et al. Melatonin versus midazolam in the premedication of anxious children attending for elective surgery under general anaesthesia: the MAGIC non-inferiority RCT. Health Technol Assess 2025;29(29). https://doi.org/10.3310/CWKF1987

Results: The trial was stopped early due to recruitment futility. Between 30 July 2019 and 9 November 2022, 110 children were recruited; 55 allocated to midazolam and 55 allocated to melatonin. Pre-planned analyses showed an adjusted mean difference of 13.1 (95% confidence interval 3.7 to 22.4) for the intention-to-treat population and 12.9 (95% confidence interval 3.1 to 22.6) for the per-protocol population, in favour of midazolam. In both analyses, the upper limit of the 95% confidence interval exceeds the predefined margin of 4.3; therefore, melatonin is not non-inferior to midazolam. The lower limit of the 95% confidence intervals excludes zero and thus melatonin is inferior to midazolam; the difference found is considered to be clinically meaningful. Adverse events in the midazolam arm (26%) were slightly higher than melatonin (18%); there were no serious adverse events in either arm. Challenges to recruitment included study-related factors (eligibility criteria and trial design), participant factors (caregiver stress on the day of treatment) and practitioner factors (valuing predictability). In terms of acceptability, preferences of the anaesthetist, patient and caregiver factors and medication side effects profile were influential and suggest the choice of preoperative anxiolytic is more complex than previously described.

On average, costs over the 14 days post surgery were lower for those who received melatonin ($-\pm46.20$, 95% confidence interval $-\pm166.14$ to ±66.74) with a mean incremental difference in procedure success of -0.02 (95% confidence interval -0.08 to 0.004), although there was uncertainty around the results.

Conclusion: In children with preoperative anxiety, midazolam is more effective than melatonin at reducing preoperative anxiety prior to general anaesthesia, although the early termination of the trial increases the likelihood of bias.

Limitations: The trial was prematurely terminated due to recruitment futility. Despite this, a clinically meaningful and statistically significant finding was observed about the primary outcome.

Future work: There remains a need to develop or repurpose another drug with a more favourable side effects profile to midazolam.

Funding: This synopsis presents independent research funded by the National Institute for Health and Care Research (NIHR) Health Technology Assessment programme as award number 16/80/08.

A plain language summary of this synopsis is available on the NIHR Journals Library Website https://doi.org/10.3310/ CWKF1987.

Introduction

Parts of this text have been reproduced with permission from Deery *et al.*¹ and Bolt *et al.*² These are Open Access articles distributed in accordance with the terms of the Creative Commons Attribution (CC BY 4.0) licence, which permits others to distribute, remix, adapt and build upon this work, for commercial use, provided the original work is properly cited. See: https://creativecommons.org/ licenses/by/4.0/. The text below includes minor additions and formatting changes to the original text.

This report details the work undertaken to establish the comparative effectiveness and side effects profile of melatonin versus midazolam for high levels of preoperative distress in children. It arose from a call commissioned by the National Institute for Health and Care Research (NIHR) Health Technology Assessment (HTA) programme, based on a Cochrane review in adults, and other systematic reviews that involved children, which suggested that melatonin may be equally as effective as standard premedication treatment with midazolam in reducing preoperative distress and anxiety as well as emergence agitation.

Rationale for research and background

There are around half a million new episodes of hospital care per year for children aged 3–14 years in the NHS

requiring general anaesthesia.³ In 2016, over a third of hospital attendances related to day case procedures.⁴ Anxiety ahead of general anaesthesia is common, with up to half of children displaying distress behaviour at the point of undergoing general anaesthetic (GA).⁵ Anxiety and distress in a child may lead to overall non-compliance and thus rescheduling of elective surgery; it may furthermore lead to greater postoperative pain, agitation and behavioural changes after surgery including sleep disturbances.⁶⁻¹⁰

Midazolam, the current standard premedication given to an anxious child ahead of surgery, has been shown to be effective,¹¹ although there are numerous adverse effects which make the medication less than ideal. One major consequence of benzodiazepine drugs such as midazolam is a sedative effect, which necessitates theatre transfer of the premedicated child on a trolley, and also significantly delays postoperative recovery;12,13 the current method of premedication therefore adds a significant burden on both resources and throughput. Further concerns relating to midazolam include the potential for respiratory suppression¹⁴ and also unpredictable effects on children which may result in agitation rather than anxiolysis particularly in children with additional needs.¹⁵ However, the degree of risk presented with midazolam is accepted due to an over-riding need for co-operation in the anaesthetic room. There is therefore a clear need to

evaluate whether there is an alternative anxiolytic to midazolam which is an effective and acceptable premedication for the management of the anxious child ahead of anaesthesia.

Melatonin has been proposed as an alternative premedication, with evidence that the drug may be as effective as midazolam at reducing preoperative anxiety in adults¹⁶⁻²² and having an excellent safety profile.^{23,24} However, studies in children have delivered conflicting results with regard to effectiveness. We presented this uncertainty in a systematic review which was unable to confirm whether melatonin was comparable in effectiveness to standard premedications, including midazolam.²⁵ Several limitations of the included studies were reported. Furthermore, whereas other trials had evaluated melatonin use in a general paediatric preoperative population, we proposed undertaking a trial in the specifically anxious population of children, which reflects standard practice of premedication use in the UK healthcare setting. The rationale was to evaluate if melatonin was non-inferior to midazolam, that is as effective as the standard-of-care premedication and with a better safety profile.

Objectives

The objectives from the trial are described below; some text in this section has been reproduced from the Melatonin for Anxiety prior to General anaesthesia In Children (MAGIC) Protocol.¹

Feasibility objectives

To undertake an internal pilot trial to determine the feasibility of a full-scale trial, in terms of:

- recruitment
- retention [adverse events (AEs) reporting and Post Hospitalisation Behaviour Questionnaire for Ambulatory Surgery (PHBQ-AS)²⁶ follow-up]
- allocation concealment and blinding.

Clinical objectives (safety and efficacy)

To evaluate if melatonin, in relation to midazolam, is:

- non-inferior in dealing with preoperative anxiety evaluated by Modified Yale Preoperative Anxiety Scale-Short Form (mYPAS-SF)^{27,28} score over the following three standard preoperative time points recommended for the scale:
 - start of transfer
 - on entry into anaesthetic room 0
 - on induction of anaesthesia 0

- superior in dealing with secondary safety and efficacy outcomes [anaesthetic turnaround time, recovery time, Paediatric Anaesthesia Emergence Delirium (PAED) scale,²⁹ Vancouver Sedation Recovery Scale (VSRS),³⁰ Revised Faces Pain Scale (FPS-R) (observer and participant reported),³¹ analgesia requirements, PHBQ-AS, AEs, orientation and cognitive/ psychomotor function]
- non-inferior in dealing with secondary efficacy outcomes (anaesthetic failure rate)
- to describe serious adverse events (SAEs) data (summarised both at patient level and event level) and report listings between the different arms.

Integrated qualitative substudy

- To inform strategies to improve recruitment, explore clinician and patient's responses to an intervention and to explain the findings of the randomised controlled trial (RCT).
- To explore stakeholder perspectives on the patient refusal of GA, acceptance of the drugs, distress reduction, impacts on recovery such as postoperative sedation and adverse effects.
- To explore patient experiences of recruitment and the acceptability of the two drugs including taste, reduction of distress, the child's postoperative recovery and any longer-term implications.

Economic objectives

Fully integrated health economic analysis to estimate the:

 cost-effectiveness of introducing melatonin, compared to usual care, over the study period and modelled to 1 year using both a cost-per-successful procedure and cost-per-quality-adjusted life-year (QALY) approach.

Methods for data collection and analysis

The MAGIC trial was a parallel-group, double-blind, multicentre, RCT to assess the non-inferiority of melatonin compared to midazolam in the treatment of anxiety in children (aged 3-14 years) undergoing scheduled, elective dental, ophthalmology, ear, nose and throat (ENT), gastroenterology, radiology, plastic, orthopaedic, urology or general surgery under GA. The trial was run across 20 UK hospital Trusts, including large Teaching Hospitals and smaller District General Hospitals. The primary outcome of the trial was the mYPAS-SF score²⁸

This synopsis should be referenced as follows: Deery C, Bolt R, Papaioannou D, Wilson M, Hyslop M, Herbert E, et al. Melatonin versus midazolam in the premedication of anxious children attending for elective surgery under general anaesthesia: the MAGIC non-inferiority RCT. Health Technol Assess 2025;29(29). https://doi.org/10.3310/CWKF1987



FIGURE 1 Melatonin for Anxiety prior to General anaesthesia In Children participant pathway. ASA, American Society of Anesthesiologists; IMP, investigational medicinal product; STAI, State–Trait Anxiety Inventory questionnaire;³³ QoL CHU9D, Quality of Life Child Health Utility 9D questionnaire.³⁴ Note: Reproduced with permission from Bolt *et al.*² This is an Open Access article distributed under the terms of the Creative Commons Attribution Licence, which permits unrestricted use, distribution and reproduction in any medium, provided the original author and source are credited. This figure includes minor additions and formatting changes to the original text.

measured preoperatively at three time points: start of transfer to theatre, entry to the anaesthetic room and induction of anaesthesia; adjusted for baseline score and other pre-planned covariates.³² The pre-defined non-inferiority margin was a 4.3-point difference between the two groups, with non-inferiority being declared if the upper bound of the 95% confidence interval (CI) did not exceed this value. The intention-to-treat (ITT) and per-protocol (PP) analysis populations were coprimary for the primary outcome.

The trial was closed early due to recruitment futility; therefore, the original sample size target of 624 participants was not reached. Instead, between 30 July 2019 and 9 November 2022, 110 children and 113 caregivers were consented and randomised into the trial; 55 children were randomised to receive melatonin and 55 to receive midazolam.

Full details of the MAGIC trial have been published as a protocol here (https://doi.org/10.15131/shef. data.22220884). *Figure 1* shows the patient pathway and data collection through the trial.

Results summary

BOX 1 Research papers synthesised in the synopsis

- Main Protocol Paper: Deery C, Bolt R, Papaioannou D, Totton N, Herbert E, Hyslop M et al. The MAGIC Trial (Melatonin for Anxiety prior to General anaesthesia In Children): A Multicentre, Parallel Randomised Controlled Trial of Melatonin Versus Midazolam in the Premedication of Anxious Children Attending for Elective Surgery Under General Anaesthesia. The University of Sheffield. Workflow; 2023. https://doi.org/10.15131/shef. data.22220884.v1
- Statistical Analysis Plan Paper: Herbert E, Totton N, Deery C, Bolt R, Hyslop M, Bradburn, M, et al. MAGIC Trial – Statistical Analysis Plan. The University of Sheffield. Workflow; 2023. https://doi.org/10.15131/shef.data.23180222
- Main Results paper: Bolt R, Hyslop MC, Herbert E, Papaioannou D, Totton N, Wilson M, et al. The MAGIC trial – a multicentre, parallel, non-inferiority, randomised controlled trial of melatonin versus midazolam in the premedication of anxious children attending for elective surgery under general anaesthesia. Br J Anaesth 2024;132:76–85. https://doi. org/10.1016/j.bja.2023.10.011. Epub November 10 2023. PMID: 37953202; PMCID: PMC10797512.
- Internal Pilot Qualitative Paper: Kettle J, Deery C, Bolt R, Papaioannou D, Marshman Z. Stakeholder perspectives on barriers and enablers to recruiting anxious children undergoing day surgery under general anaesthetic: a qualitative internal pilot study of the MAGIC randomised controlled trial. *Trials* 2021;22:458. https://doi.org/10.1186/s13063-021-05425-z. PMID: 34271982; PMCID: PMC8285773.

- Lessons Learned Paper: Hyslop MC, Papaioannou D, Bolt R, Wilson M, Bradburn M, Clarkson J, *et al.* Barriers and enablers to recruiting participants within perioperative and anaesthetic settings: lessons learned from the MAGIC trial. *BJA Online* 2025;**13**:100375.
- Main Qualitative Paper: Kettle J, Bolt R, Deery C, Papaioannou D, Rodd H, Hyslop MC, et al. Acceptability of midazolam and melatonin as premedications for anxious children undergoing general anaesthetic: a qualitative interview study with children, caregivers and health professionals participating in the MAGIC randomised controlled trial. *Trials* 2024;25:813. https://doi.org/10.1186/s13063-024-08611-x
- Health Economics Paper: Young T, Papaioannou D, Bolt R, Herbert E, Totton N, Hyslop MC, et al. The MAGIC trial (Melatonin for Anxiety prior to General anaesthesia In Children) – Health Economics Report. In progress of submission.
- Study within a trial Paper: Herbert E, Papaioannou D, Loban A, Totton N, Hyslop M, Bolt R, Deery C. Personalised versus standard text message prompts for increasing trial participant response to telephone follow-up: an embedded randomised controlled retention trial. *Trials* 2024;25:108. https://doi. org/10.1186/s13063-024-07916-1

Recruitment futility

In November 2022, it was agreed with the Trial Steering Committee (TSC) and the NIHR (trial funder) that the MAGIC trial would be terminated prematurely on account of recruitment futility. *Figure 2* outlines the timeline of the MAGIC trial from opening to closure (2019–22). The trial recruited 110 participants out of the target 624, with 568 participants screened. The trial opened to recruitment in July 2019, experiencing slower than expected recruitment during its pilot phase with summer holidays affecting staff capacity and availability. Although recruitment steadily increased up to December 2020, several recruitment barriers were identified, which included:

- reduced number of children requiring premedication prior to surgery
- eligibility restrictions such as limited surgical specialties
- 24-hour ward admission protocols at some sites and younger children (3- to 4-year-olds) who were previously ineligible
- very anxious children requiring more than one premedication, for example patients with high anxiety and autism
- anaesthetist equipoise regarding melatonin use and increasing use of other premedications instead of midazolam (e.g. dexmedetomidine)
- delays on theatre lists and waiting times due to potential delays in dispensing investigational medicinal product (IMP)
- research pharmacy opening times preventing anxious children at the beginning of theatre lists from being recruited
- language barriers
- patients with high anxiety unable to provide assent or declining involvement in the trial

• research staff availability and capacity.

Barriers were discussed and changes made to the trial protocol to try and alleviate these issues (see Appendix 1, Table 2 - Protocol changes) and included: expansion of inclusion criteria to allow 3- and 4-year-old children to be recruited and include more surgical specialties: gastroenterology, radiology, plastic, orthopaedic. urology or other general surgery; change to clarify assent requirement from children, with children who neither provide assent nor decline the trial (due to high anxiety) able to be enrolled based on caregiver consent and principal investigator (PI) decision (children who verbally decline to participate not be included); change to allow local care teams to send out study information to potential participants prior to their day of surgery; and change to allow children the option of reviewing the information sheet or the video and not a requirement to undertake both.

In early 2020, the coronavirus pandemic led to numerous pressures and impacted the NHS as well as research. The MAGIC trial was particularly vulnerable: routine elective surgeries were halted during the pandemic; PIs were primarily anaesthetists essential in the pandemic response for the care of severely ill COVID-19 patients on mechanical ventilators. Midazolam became subject to nationally controlled use as the drug was favoured as first-line sedation for patients requiring mechanical ventilation and our manufacturer could not source the product for trial supplies. Brexit further impacted the trial due to European Union (EU) medicinal drug imports being subject to new legislation and causing large delays to the manufacture of trial drug supplies.

The pandemic and Brexit had profound effects on trial recruitment, and recruitment was suspended on 27 March 2020. Although recruitment to the MAGIC trial recommenced on 9 October 2020, the pandemic further hampered the trial's progress with site capabilities to recruit and treat participants affected by cancellation of elective surgeries, isolation of surgical anaesthetic teams with COVID-19, and redeployment of research nurses and clinical staff to COVID-19 studies. Recruitment to the trial was stop/start between October 2020 and November 2022 due to these issues. Despite major efforts from site research teams, protocol amendments to address recruitment barriers, and genuine PI and research staff enthusiasm for the trial, there was not sufficient evidence that site participation was translating to recruits for the team to make a cogent case to continue. The changes made to the Protocol, as part of substantial amendment 5, were also insufficient to improve recruitment significantly;

This synopsis should be referenced as follows:

Deery C, Bolt R, Papaioannou D, Wilson M, Hyslop M, Herbert E, et al. Melatonin versus midazolam in the premedication of anxious children attending for elective surgery under general anaesthesia: the MAGIC non-inferiority RCT. Health Technol Assess 2025;29(29). https://doi.org/10.3310/CWKF1987



FIGURE 2 Melatonin for Anxiety prior to General anaesthesia In Children trial timeline. DMEC, Data Monitoring and Ethics Committee; TMG, Trial Management Group.

for example expansion of surgical specialties to gastroenterology, radiology, plastic, orthopaedic, urology or other general surgery only yielded four patients from those new specialties included. The last trial participant was recruited and followed up by 23 November 2022.

Primary and secondary outcomes in the Melatonin for Anxiety prior to General anaesthesia In Children trial

Results of the MAGIC trial are reported in Bolt *et al.*² A detailed full version of the MAGIC statistical analysis plan is also available (https://doi.org/10.15131/shef. data.23180222).³⁴ *Box* 1 highlights the relevant research outputs generated from the MAGIC trial.

Primary outcome – modified yale preoperative anxiety scale-short form

Despite the early termination of the trial, the planned analyses as defined in the statistical analysis plan were completed as appropriate to sufficient data being available. Due to not achieving the original sample size target, *p*-values have not been reported and, instead, the presentation has focused on CIs.

The primary outcome, mYPAS-SF, represents preoperative anxiety with higher values representing increased anxiety. In the ITT analysis population, the adjusted mYPAS-SF score in the melatonin group was higher than in the midazolam group (adjusted mean difference 13.09, 95% CI 3.74 to 22.44; n = 92). Similar results were observed in the PP population (adjusted mean difference 12.9, 95% CI 3.1 to 22.6; n = 87). Melatonin was shown to be inferior to midazolam reducing preoperative anxiety in both analysis populations.

Secondary outcomes

Secondary outcomes showed small numbers of anaesthetic failure in the trial (two in the melatonin group and one in the midazolam group). Anaesthetic turnaround and recovery times were similar in the two groups [adjusted ratios of means 1.01 (95% CI 0.78 to 1.3) and 0.88 (95% CI 0.74 to 1.04), respectively]. Analgesia requirements were also similar between the groups [adjusted odds ratio (OR) 1.22, 95% CI 0.37 to 4.25]. Longitudinal analyses of repeated recovery scores (VSRS, co-operation score, FPS-R and PAED) also showed no differences between the groups² (see *Appendix 2, Table 3*).

Fourteen-day follow-up

At 14 days post surgery, the primary caregiver was contacted to complete a PHBQ-AS form to assess postoperative behavioural changes. The results of these questionnaires showed no difference between the groups (adjusted mean difference -0.03, 95% CI -0.21 to 0.14; n = 74). There were 11 AEs in the melatonin group among 9 of the 49 patients (18.4%, 95% CI 9.2% to 32.5%), and 23 AEs in the midazolam group among 13 of the 50 patients (26.0%, 95% CI 15.1% to 40.6%). No SAEs were reported during the trial. Using these trial results, we concluded that melatonin is inferior to midazolam at

reducing preoperative anxiety in children who require premedication and there appeared to be no impact on the success of the procedure or recovery outcomes.

Qualitative summary

An integrated qualitative substudy was undertaken to: (1) explore health professional, and patient (child) and caregiver experiences of recruitment during an internal pilot to inform recruitment strategies; (2) explore healthcare professional and patient experiences on the acceptability of the two drugs; and (3) explain the findings of the RCT.

During the internal pilot, a qualitative interview study was undertaken³⁵ to identify research staff perspectives on barriers and enablers to recruitment. Sixteen stakeholders including research nurses, PIs and other investigators from six trial sites participated in semistructured interviews; data were analysed in framework analysis. Barriers and facilitators related to various aspects of MAGIC, including the study itself, the potential participants and their caregivers, practitioners (and those with whom they are collaborating outside of the study, such as other clinicians and nurses) and wider setting and contextual factors. A key study-related barrier was the eligibility criteria, which was broadened during the course of the study. Barriers related to patients and caregivers included lower than expected numbers of eligible patients, and the impact of the child's anxiety on the caregivers' willingness to consider MAGIC and focus on recruitment materials. Anaesthetists and research nurses involved in MAGIC were not always available when potentially eligible patients were scheduled for treatment, while anaesthetists outside of MAGIC did not always give permission for their patients to be recruited, for example if they had preferences for other premedications. The high-pressure, fast-paced nature of the surgical day unit setting could also be a barrier, as could the hospital context, for instance there were frequent problems obtaining the trial drugs from the hospital pharmacy in time for patients at the start of the theatre list.

The acceptability of midazolam and melatonin as premedications was explored in the qualitative interview study with children, caregivers and healthcare professionals. Thirty-seven participants (23 health professionals, 10 caregivers and 4 children) were interviewed. Factors around the acceptability of melatonin and midazolam were identified. These included: effectiveness as premedication prior to a GA; administration of premedication; experience of recovery from GA; prior experiences of premedication; and associations and evidence of the range of available options for managing anxiety. Factors around the acceptability of premedications more generally were also identified and it was noted that decisions about premedications appear to be made based on several more complex factors than merely anxiolytic properties.

Health economic summary

An economic evaluation was undertaken alongside the MAGIC trial.³⁶ The aim of the health economic analysis was to evaluate the within-trial cost-effectiveness of melatonin for anxiety in children compared to usual care (midazolam) prior to general anaesthesia in children from an NHS and Personal Social Services perspective. Resource use was collected from case-report forms from randomisation to 14 days post surgery. The main outcome was the incremental cost-per-successful procedure with a secondary outcome of incremental cost per QALYs with QALYs measured using the Child Health Utility-9D (CHU9D). QALYs were not selected as the primary analysis due to the short (14 day) time frame of the study and the lack of evidence that QALYs would change over this time frame and due to not having a validated measure of QALYs in children across the age range of the study.

Of the 110 participants recruited, a total of 100 children had the intervention drugs prescribed, 50 in the melatonin arm and 50 in the midazolam arm. Of these, one child in the melatonin arm refused to take the drug. These 100 children were the focus of the health economic analysis. On average, costs over 14 days were lower for those who received melatonin ($-\pounds46.20$, 95% CI $-\pounds166.14$ to $\pounds66.74$) with a mean incremental difference in procedure success of -0.02 (95% CI -0.08 to 0.004), although there was uncertainty around the results. When the incremental cost-effectiveness ratios cover more than one quadrant of the cost-effectiveness plane, it is recommended that 95% CIs are not reported owing to issues being able to distinguish between situations when the new treatment is favoured and when it is not.³⁷

The study closed early owing to issues with recruitment and this has limited the economic analysis; subgroup analysis was limited to those who underwent head and neck procedures owing to small numbers by subgroup for other procedure types and age groups. For head and neck procedures (melatonin n = 46; midazolam n = 47), the results were similar to the primary cost-effectiveness analysis, on average melatonin was cheaper (mean £45.20, 95% CI -£170.39 to £76.37) with a mean incremental difference in procedure success of -0.022 (95% CI -0.09

Deery C, Bolt R, Papaioannou D, Wilson M, Hyslop M, Herbert E, et al. Melatonin versus midazolam in the premedication of anxious children attending for elective surgery under general anaesthesia: the MAGIC non-inferiority RCT. Health Technol Assess 2025;29(29). https://doi.org/10.3310/CWKF1987

to 0.04). The cost-utility analysis for the full sample used CHU9D to estimate utilities and was carried out after imputing missing CHU9D (43% missing data) and also produced uncertain results. The mean difference in costs remained the same as for the cost-effectiveness analysis (-£46.20, 95% CI -£166.14 to £66.74) with a mean incremental QALY 0.0 (95% CI -0.0008 to 0.0008).

Discussion and interpretation

The MAGIC trial is the first multicentred RCT comparing the effectiveness of midazolam compared to melatonin in anxious children, as a premedication prior to general anaesthesia. Despite not recruiting to the sample size target, the primary outcome in both the ITT and PP analyses showed that midazolam was more effective as a premedication than melatonin on preoperative anxiety. This was the case due to a larger effect size than expected being identified between the two treatments. This result differs from the findings of previous studies.^{38,39} This may be because the previous studies had smaller sample sizes and used a sample of all available patients rather than targeting anxious patients. Therefore, previous studies are likely to be underpowered, and, as a result of including non-anxious individuals, diluted in the observation of any differences in effectiveness between the two drugs.

Regarding the secondary outcomes, no differences between the two drugs were seen. Only three patients (n = 2 in the melatonin arm and n = 1 in the midazolamarm) did not have their planned procedure due to failure to be anaesthetised. This suggests, as would be expected, that both drugs were having an anxiolytic effect. Analgesia requirements were also similar between the groups. Longitudinal analyses of repeated recovery scores (VSRS, co-operation score, FPS-R and PAED) also showed no evidence of difference between the groups. Due to the sedative effects of midazolam, it was predicted that readiness for discharge time would be less in the melatonin arm, although this was not observed.⁴⁰ It may be that any differences in secondary outcomes were not detected due to lack of power. At the 14-day follow-up, no differences were found between the two arms (PHBQ-AS). There were no SAEs in either arm. There were slightly more participants with at least one AE in the midazolam arm (13) compared to the melatonin arm (11). However, only one of these in the midazolam arm was thought to be potentially related to the premedication. Both drugs demonstrated excellent safety profiles.

A qualitative semistructured interview study was undertaken to identify facilitators and barriers to

recruitment. The sample included research nurses, PIs and other investigators from six trial sites. The finding that there were fewer eligible patients than expected is frequently reported from other trials.^{41,42} For the MAGIC trial, all potential sites conducted an audit prior to the trial to confirm an adequate throughput of eligible patients; however, despite this, the numbers of eligible patients fell below what was predicted.

One barrier to recruitment was the large amount of information caregivers had to digest at the time of the child's surgery, particularly when they may have been wishing to focus solely on their anxious child. Also, the caregiver may understandably be anxious themselves on the day of surgery. The decision when designing the trial to have recruitment on the day was to mirror current practice where the final decision about the need for a premedication is made by the anaesthetist responsible for the patient on the day. In response to this feedback, MAGIC did look to address this by recruiting patients at pre-assessment clinics; however, this protocol amendment was never instigated due to the early closure of the trial. Interestingly, one of the enablers to recruitment was a desire expressed by caregivers to 'give back to the NHS' by participating in research. As with other trials, eligibility criteria such as patient age (so that validated measures can be used) were a barrier to recruitment.⁴³ As is often the case, time constraints on key members of staff, in what is a pressured and time-sensitive environment, was a further barrier to recruitment.43

On average, melatonin was £14.20 less per patient over the 14-day follow-up period, although there was uncertainty in the results and the difference in effectiveness between the two drugs questions the value of this saving. This is the first study to report on the costeffectiveness of melatonin compared with midazolam in children, young people and adults. Studies that have looked at the cost-effectiveness of midazolam have not been conclusive.⁴⁴⁻⁴⁷

A strength of the current study is that it is the most relevant study to inform practice in the UK NHS. It is also very strong methodologically. Patients, their caregivers and all staff delivering care, including the anaesthetists, were blind to the intervention given. The chosen outcome measure is well validated and recognised; furthermore, mYPAS-SF is applicable both in the preoperative area as well as during induction. However, a major weakness is that the target sample size was not achieved, and although a clear statistical and clinically meaningful difference in favour of midazolam was found, as stated there may not have been an adequate sample size to identify differences in the secondary outcomes. Additionally, results from under-recruited RCTs can require caution due to large variation or bias in the sample achieved. Although the use of mYPAS-SF is a strength, other common outcome measures, such as mask acceptance or level of restraint, were not used.48,49

Many trials were affected by the COVID-19 pandemic.⁵⁰ MAGIC was particularly vulnerable to the impact as all routine elective surgeries halted during the pandemic; Pls were mainly anaesthetists essential in the pandemic response for the clinical care of critically ill COVID-19 patients; midazolam was favoured as first-line sedation for patients requiring mechanical ventilation. Brexit further impacted the trial due to EU medicinal drug imports being subject to new legislation which caused large delays to the manufacture of trial drug supplies.⁵¹ A number of other challenges for recruitment included trial staff resources and availability, particularly research nurse availability, and research pharmacy opening times which meant the IMP was not available prior to the beginning of morning theatre lists as for clinical reasons, anxious children are usually placed first on a list if possible. There were also issues around anaesthetist and other staff's equipoise. resulting in an unwillingness to recruit anxious children or children with special needs.

Increased use of dexmedetomidine as a premedication in children, among UK anaesthetists, which gathered pace over the chronology of the MAGIC trial, was a further barrier to recruitment at some sites. Anecdotally, the drug is seen as better tolerated and potentially more effective as a preoperative anxiolytic, it is administered intranasally as opposed to orally, eliminating the risk of the patient spitting out the medication, and it has a superior side effects profile to midazolam.⁵² However, to date, none of the studies directly comparing the drugs have confirmed dexmedetomidine's superiority as a premedication in children over midazolam.

Stakeholders were engaged through a number of patient and public involvement (PPI) groups run during the trial design process. Advice was provided from child and parent groups with respect to the wording of patient information leaflets, as well as the design and wording of an age-specific trial video. Oral and dental PPI groups provided further advice on trial design so as to ensure the patient care pathway was not adversely affected by the trial. Anaesthetist expert groups were engaged at the beginning to establish an agreed minimal clinically important difference, which informed the sample size calculations used in the trial.

All research teams were trained in mYPAS analysis using an online resource accessed through a collaboration with researchers running the 'Little Journey' trial (www. littlejourney.health/). An online mYPAS training resource shared by the Little Journey team allowed interexaminer reliability to be maintained at a κ of 0.7. This activity strengthened the available pool of research nurses available to undertake trial observations throughout recruitment, thereby improving the overall capacity for the trial.

Challenges exist in conducting perioperative trials, including operational difficulties such as research pharmacy opening times and wide variation in anaesthetist practice. In practice, decisions about premedications are taken based on the balance of several complex issues. Other relevant measures of sedation that were not used in this trial, such as mask acceptance or level of restraint, may have provided further insight into the overall effects of each of the trial medications. Wide CIs of secondary outcome measures limit their interpretation including factors not based purely on their anxiolytic properties.

Patient and public involvement

Aim

The aim of PPI in all aspects of the study was to ensure that the views of children, their caregivers and their wider families were dispersed through the research, such that the trial was acceptable, and the results would be of direct benefit to them.

Methods

We worked with PPI representatives from grant preparation through to dissemination. The study arose from a commissioned call, and we worked with child and parent representatives to further support the design and provision of information. During the design of the trial, children who had a recent GA attended a study-specific PPI meeting along with their parents.

Involvement from the PPI group was provided during the development and iteration of participant information resources - including a participant video. PPI representatives were involved in research management and troubleshooting as members of the Trial Management Group (TMG) and TSC. PPI input was sought for the interpretation of the quantitative and qualitative data, writing and dissemination of the findings to the trial participants and their caregivers.

Results of patient and public involvement input

During grant preparation, the PPI group considered the importance of the study and felt that the potential side

This synopsis should be referenced as follows: Deery C, Bolt R, Papaioannou D, Wilson M, Hyslop M, Herbert E, et al. Melatonin versus midazolam in the premedication of anxious children attending for elective surgery under general anaesthesia: the MAGIC non-inferiority RCT. Health Technol Assess 2025;29(29). https://doi.org/10.3310/CWKF1987

effects of midazolam were a strong reason to look for an alternative drug. There was unanimous agreement that measuring the ability of each drug to 'calm the child' was the most important aspect to assess, followed by the safety profile of each drug. The mYPAS-SF was discussed, and parents felt reassured to have an extra nurse in the anaesthetic room measuring this, rather than relying on the recovery nurse. Parents also felt an objective measure assessed by a researcher was more appropriate than asking the child. All children described how they would feel the same about having a GA, regardless of the surgical procedure, for example dental extractions or tonsillectomy. The PPI group suggested follow-up should be done at 2 weeks postoperatively by a single contact, as children who undergo tonsillectomy remain housebound for 2 weeks. Telephone contact was preferred for follow-up, the group discounted e-mail and advised a text message reminder.

During the trial, the PPI provided advice to improve the lay summary and assisted in the design of the interview schedules for the qualitative research. They participated in the design of the parent and child information leaflets and children's information videos to ensure the information was age-appropriate. Examples of how PPI shaped the research included consideration of how to promote recruitment when it was slower than anticipated (e.g. through relevant networks). PPI members sat on the TSC and TMG for the trial. This ensured the research was monitored in an equitable manner, particularly with respect to trial progression criteria. This also allowed PPI views on the proposed amendments to the trial procedures to improve recruitment.

Patient and public involvement members made suggestions to improve materials, which reported the trial results to ensure the needs of the service user audience were met, in particular, that the materials were age-appropriate.

Discussion of patient and public involvement input

Patient and public involvement input was key for the design of MAGIC, tailoring it to the relevant audiences and leading discussions around what would and would not work. It also helped shape the appropriate dissemination to relevant audiences.

Reflections and critical perspective

The study had the involvement of representatives with lived experience, but we noted that further input may have been helpful from those from minority ethnic groups, or those with autism, especially to address issues around language and other barriers to joining the trial as seen during the pilot phase and qualitative review.³⁵

Equality, diversity and inclusion

The MAGIC trial enrolled 110 anxious paediatric patients (aged 3–14 years) undergoing GA prior to surgery, and their primary caregivers, from 20 UK hospitals sites. The mean age of participants was 7.9 years [standard deviation (SD) 2.6], with 48% male sex and 52% female sex. A total of 81% participants were American Society of Anesthesiologists (ASA) Classification System I status and 18% were ASA II status. ASA III–V was not included due to the comorbidities increasing the risk of GA. The study did include patients with neurodiversity and learning difficulties; however, there is not a breakdown of the proportions of this.

A total of 93% of the trial participants were White English, Welsh, Scottish, Northern Irish, British; 1% Indian; 1% Pakistani; 1% white and Black Caribbean; 1% white and Asian: 1% any other white background and 1% African - or 94% white ethnic background and 5% were from a non-white ethnic background. The other 1% is missing data. In Office for National Statistics data on live births in 2020 (the most recent year of recruitment into the MAGIC trial with data available), 70% of children were from a white ethnic group, and 26% of children were from a non-white ethnic group (and 4% not stated), suggesting that children within MAGIC were not fully representative of the wider population. The majority of the trial participants (66%) were undergoing dental surgery. A review by Levine⁵³ on childhood caries and hospital admissions in 5-year-old children showed a higher prevalence of caries in 'other ethnic group' (44.3%) and Asian/Asian British ethnic groups (36.9%) than that found in other ethnic groups and other minority ethnic children and compared to those classed as 'white'. This highlights some discrepancies with our ethnicity proportions within MAGIC participants.

The Official Statistics Hospital tooth extractions (0- to 19-year-olds) 2021 stated a clear gradient in the most deprived areas having the highest rates of caries-related tooth extractions. A wide range of hospitals from diverse geographical areas were included within the trial; this included District General Hospitals as well as specialist Children's Units, across the UK, in an attempt to broaden recruitment possibilities. The trial also included hospitals centred within more deprived areas such as Doncaster, Barnsley, Middlesbrough, Sunderland, Bolton, Glasgow, Fife, Kilmarnock and others.

As part of the PPI input into the MAGIC trial, children and parents from representative demographics were included in the design of the trial to ensure reflective and inclusive views were incorporated, including children who had recently undergone ENT and dental surgeries under general anaesthesia. A video was suggested (and devised) to support provision of trial information in a more easily understandable and manageable format. On completion, the video was presented to PPI groups and received positive feedback. However, it was noted as part of the pilot and qualitative review that language barriers remained an issue for inclusion, and future trials within this research area need to consider understanding and access to trials for children whose first language is not English.

Impact and learning

The trial results demonstrated that midazolam, the current standard of care, is the most effective treatment available; therefore, there is no immediate requirement for a change in clinical practice. Midazolam remains a safe, effective drug in the management of preoperative anxiety in children. However, we anticipate that in the longer term, further research will be required to identify an alternative premedication with an improved side effects profile. While there are other candidate anxiolytic drugs which could be evaluated, the trial findings remove melatonin from consideration as a care standard.

Lessons learnt

The MAGIC trial experienced numerous delays and challenges during its run from 2019 to 2022 as discussed in the *Recruitment futility* section. Many of the challenges to recruitment were those often experienced by RCTs; however, some issues encountered were specific to the anaesthesia and perioperative medicine setting. The 'Barriers and enablers to recruiting participants within perioperative and anaesthetic settings: Lessons learnt from the MAGIC trial'⁵⁴ paper gives an overall account of the individual challenges experienced during the trial, and potential solutions proposed to mitigate said issues, *Table 1*, taken from the article, highlight and summarise these accordingly.

TABLE 1 Melatonin for Anxiety prior to General anaesthesia In Children lessons learnt summary table

Issue encountered	Strategies to overcome	Challenge remains
Trial staff resources and availability		
Research pharmacy opening times		
• Opening times of research pharmacy were 1-2 hours later than when the surgical list commenced. Anxious patients were often placed at the beginning of surgical lists	 Whenever possible, allow consent and eligibility to be completed at pre-assessment clinics prior to surgery (eligibility may need to be reconfirmed on the day or surgery) Consider randomisation prior to day of surgery Consider if IMPs can be held and dispensed outside of research pharmacy (would need a risk assessment with regard to IMP accountability) Consider if research pharmacy can be costed and staffed outside of normal hours (would need costing into trial grant applications) 	before the day itself
Anaesthetist trainee network		
 Trainee networks uncontactable or lack of engagement 	 Consider barriers and enablers for using trainee networks in the perioperative setting. An alter- native model for Anaesthetic Trainee Networks required for effective engagement Consider individual trainees, with commitment to academic experience, assigned to involve- ment with trial for an extended period, to champion colleague engagement 	Organisation and commitment o each network continue to vary
Associate PI scheme		
 Trainee rotations are 3–6 monthly and preclude involvement in the associate PI scheme 	• Consider timelines of the associate PI scheme (currently minimum 6 months) and how this fits in with trainee rotations (3 months)	• Six-month minimum requirement for participation still required
		continued

This synopsis should be referenced as follows:

Deery C, Bolt R, Papaioannou D, Wilson M, Hyslop M, Herbert E, et al. Melatonin versus midazolam in the premedication of anxious children attending for elective surgery under general anaesthesia: the MAGIC non-inferiority RCT. Health Technol Assess 2025;29(29). https://doi.org/10.3310/CWKF1987

Issue encountered Strategies to overcome **Challenge remains** Recruitment projections Limited staff availability Consider higher scrutiny of feasibility forms and Staff support and availability Overoptimistic recruitment projections keeping recruitment projections realistic continue to be an issue across Contact local Clinical Research Networks to clinical trials query whether support staff are available (for NIHR portfolio trials) Anaesthetist equipoise Multifactorial choice of premedication Consider complexities of trials The decision on the choice of premedication in Discrete choice experiments may be required children is multifactorial and not limited purely in the planning stages of future premedication that are multidisciplinary in to its anxiolytic properties. Can involve: trials to determine the attributes of premedinature, that is those that require Pharmacological effects of the drug cations important to healthcare professionals, input from various clinical disci-Clinical features and comorbidities of child patients and their caregivers plines, and involve multifactorial Palatability decision-making when deciding Child acceptance of drug route a treatment pathway • Anaesthetist equipoise and variation in clinical practice Local policies on prescribing practice (e.g. Consider undertaking a survey to understand Variation in the degree of antwo premedications) the clinical practice of anaesthetists across a aesthetist equipoise remains an Large variation in prescribing practice among large number and types of sites (i.e. Teaching ongoing issue within anaesthesia anaesthetists (e.g. use of dexmedetomidine) Hospitals, District General Hospitals, etc.) and perioperative trials Site assessment templates designed to capture Disrupted equipoise among anaesthetists regarding interventions and ensure consistent prescribing practices for anaesthetists across sites and within sites (i.e. survey more than one anaesthetist per site) New treatments becoming available Consider the potential for the treatment land-Treatment landscapes will For example, dexmedetomidine scape to vary over the lifecycle of a trial continue to evolve during the Again, consider undertaking a survey to underlifetime of a clinical trial stand clinical practice among anaesthetists and whether newer 'off-label' drugs are/will be used Dual premedication use for some subgroups Site-level clinical variations may Several hospitals give two premedications to Identify local practices during feasibility or wid-• children as standard for preoperative anxiety. ening inclusion criteria where possible continue to be an issue within Particularly children with additional needs clinical trials. Exploring these at the earliest opportunity at the trial design stage is vital Equality, diversion and inclusivity Willingness to randomise some individuals, Consider greater staff education on inclusion. Inclusivity in trials is a key for example neurodiverse children or those Children with special needs form a large part of research priority. Key challenges with learning difficulties (experimental treatthose requiring premedications, particularly those which remain are ensuring: ment) within the dental setting. Research staff need to A more diverse research workbe aware that inclusion of these children is vital, as force and recruiters within trials they represent a significant part of the population More inclusive PPI representation within trials and thus deserve representation also **Recruitment setting** Day of surgery Time pressure to consent and randomise Allow flexibility to randomise before the day of There will be occasions within surgery wherever possible, while being mindful patients all on the day of surgery, in order to the perioperative setting where commence surgical lists on time, in the midst for the potential of post-randomisation droprandomisation will have to be on of other time pressures outs the day, or even at the time of Time pressure for pharmacy to blind and Consider using a range of staff to recruit surgery, which requires facilitadispense IMP prior to surgery patients, for example research nurses, where tion possible

TABLE 1 Melatonin for Anxiety prior to General anaesthesia In Children lessons learnt summary table (continued)

TABLE 1 Melatonin for Anxiety prior to General anaesthesia In Children lessons learnt summary table (continued)

Issue encountered	Strategies to overcome	Challenge remains
Paediatric assent		
 Requirement for assent even in children aged 5–7 years old Made more challenging with the requirement to assent an already anxious population 	 Consider the practicalities of assenting younger and/or anxious children, and whether it may be reasonable to seek parental consent only in certain situations PPI is crucial for input on the acceptability of including assent 	 Assent continues to be recom- mended when recruiting children to clinical trials. This needs to be considered against the practicali ties on a trial-by-trial basis

Note

Reproduced with permission from Hyslop et al.⁵⁴ This is an Open Access article distributed under the terms of the Creative Commons Attribution Licence, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited. This table includes minor additions and formatting changes to the original text.

Study within a trial: Melatonin for Anxiety prior to General anaesthesia In Children

Given the prevalence of the use of RCT, there is an ongoing need to develop and evaluate strategies for improving retention to increase efficiency. An assessment of optimum strategies can be completed by embedding a smaller RCT to evaluate these strategies within a reallife host trial, known as a study within a trial (SWAT).⁵⁵ In MAGIC, a SWAT was undertaken to evaluate the effectiveness of a personalised text message, including the recipient's first name, versus a standard text message, for prompting response in caregivers to answer and complete the 14-day telephone follow-up questionnaires within the trial.⁵⁶ The SWAT showed some evidence that personalised text messages can be effective at increasing response rates when data are collected via telephone. The intervention is low cost and practical to implement, suggesting further use and investigation within trials when scheduled telephone follow-ups are being used.

Dissemination plans

The findings from the MAGIC trial have been reported in several academic publications. Please see the Publications section for a list of outputs from the trial. The main trial results will be disseminated at Anaesthesia Research 2023 (Royal College of Anaesthetists) conference scheduled in York for 28 November 2023. The results of the trial were provided to child participants and their caregivers via postal leaflets sent out in September 2023. Our TSC/TMG PPI representatives assisted in developing a lay summary of the study results as well as reviewing the patient dissemination documents.

Links to conference presentations, publications and lay summaries will be posted on the Sheffield Clinical Trials Research Unit Twitter account and the MAGIC website.

Implications for decision-makers

This study adds to the body of literature on melatonin use by providing a definitive answer in an area where uncertainty remained. Although the study was terminated early for recruitment futility, the result was clinically meaningful and statistically significant (evidenced by the exclusion of zero in the 95% CI). The results show that melatonin is inferior to midazolam, and midazolam should remain the standard of care for premedication of anxious children ahead of surgery. However further research is needed into sourcing an alternative premedication with an improved side effects profile.

Research recommendations

We identified the following questions for future research and have indicated the area of research to which they relate.

Further interventions

- What other interventions may be effective in reducing preoperative anxiety in children?
- Is dexmedetomidine (or alternatives) a viable substitute premedication to midazolam?

Stakeholder views on important attributes of premedication

 What do stakeholders, patients and caregivers consider to be the most important in the attributes of a premedication?

Non-pharmacological interventions

 What is the clinical effectiveness and cost-effectiveness of using non-pharmacological interventions, on their own, and in conjunction with pharmacological interventions in the reduction of preoperative anxiety?

This synopsis should be referenced as follows: Deery C, Bolt R, Papaioannou D, Wilson M, Hyslop M, Herbert E, et al. Melatonin versus midazolam in the premedication of anxious children attending for elective surgery under general anaesthesia: the MAGIC non-inferiority RCT. Health Technol Assess 2025;29(29). https://doi.org/10.3310/CWKF1987

Conclusions

The trial did not reach the required sample size and therefore is prone to bias. In children with preoperative anxiety, midazolam is more effective than melatonin at reducing preoperative anxiety prior to general anaesthesia. Challenges exist in conducting perioperative trials including operational difficulties, such as research pharmacy opening times and wide variation in anaesthetist practice. There remains a clinical need to develop or repurpose another premedication with a more favourable side effects profile. In practice, decisions about premedications are taken based on the balance of several complex factors and not based purely on premedication anxiolytic properties. Future studies evaluating premedications in children may need to consider the use of discrete choice experiments to understand the preferences of children caregivers and healthcare professionals on the attributes of premedications.

Additional information

CRediT contribution statement

Christopher Deery (https://orcid.org/0000-0001-7526-7736): Conceptualisation, Funding acquisition, Methodology, Supervision, Visualisation, Writing – original draft, Writing – reviewing and editing.

Robert Bolt (https://orcid.org/0009-0007-1705-8400): Conceptualisation, Funding acquisition, Methodology, Supervision, Writing – original draft, Writing – reviewing and editing.

Diana Papaioannou (https://orcid.org/0000-0002-6259-0822): Conceptualisation, Funding acquisition, Methodology, Project administration, Supervision, Writing – original draft, Writing – reviewing and editing.

Matthew Wilson (https://orcid.org/0000-0002-9704-5189): Conceptualisation, Funding acquisition, Methodology, Supervision, Writing – original draft, Writing – reviewing and editing.

Marie Hyslop (https://orcid.org/0009-0005-7078-4933)**:** Data curation, Investigation, Project administration, Visualisation, Writing – original draft, Writing – reviewing and editing.

Esther Herbert (https://orcid.org/0000-0002-1224-5457): Data curation, Formal analysis, Validation, Writing – original draft, Writing – reviewing and editing.

Nikki Totton (https://orcid.org/0000-0002-1900-2773): Formal analysis, Methodology, Validation, Writing – original draft, Writing – reviewing and editing.

Zoe Marshman (https://orcid.org/0000-0003-0943-9637): Conceptualisation, Funding acquisition, Methodology, Writing – reviewing and editing.

Tracey Young (https://orcid.org/0000-0001-8467-0471**):** Conceptualisation, Formal analysis, Funding acquisition, Methodology, Writing – reviewing and editing.

Jennifer Kettle (https://orcid.org/0000-0002-2776-1243)**:** Formal analysis, Investigation, Writing – reviewing and editing.

Sondos Albadri (https://orcid.org/0000-0001-6516-9497): Writing – reviewing and editing.

Simon Atkins (https://orcid.org/0000-0003-3806-0968): Conceptualisation, Funding acquisition, Methodology, Writing – reviewing and editing.

KatieBiggs(https://orcid.org/0000-0003-4468-7417):Supervision, Writing – reviewing and editing.

Janet Clarkson (https://orcid.org/0000-0001-5940-2926): Conceptualisation, Funding acquisition, Methodology, Writing – reviewing and editing.

Chris Evans (https://orcid.org/0000-0001-8586-9618**):** Writing – reviewing and editing.

Laura Flight (https://orcid.org/0000-0002-9569-8290): Writing – reviewing and editing.

Jacqui Gath (https://orcid.org/0000-0001-8875-5894): Writing – reviewing and editing, PPI.

Fiona Gilchrist (https://orcid.org/0000-0002-0418-6274**):** Conceptualisation, Funding acquisition, Methodology, Writing – reviewing and editing.

Kate Hutchence (https://orcid.org/0000-0001-7870-2844**):** Data curation, Investigation, Visualisation, Writing – reviewing and editing.

Nicholas Ireland (https://orcid.org/0000-0002-3665-0641): Writing – reviewing and editing.

Amanda Loban (https://orcid.org/0000-0001-7089-1580): Data curation, Writing – reviewing and editing.

Amy Norrington (https://orcid.org/0009-0003-4854-0006): Writing – reviewing and editing.

Hamish Paton: Writing - reviewing and editing.

Jaydip Ray (https://orcid.org/0000-0001-7421-1263): Conceptualisation, Funding acquisition, Methodology, Writing – reviewing and editing.

Helen Rodd (https://orcid.org/0000-0003-2973-2558)**:** Conceptualisation, Funding acquisition, Methodology, Writing – reviewing and editing.

Elena Sheldon (https://orcid.org/0000-0002-1927-471X**):** Data curation, Investigation, Writing – reviewing and editing.

Richard Simmonds (https://orcid.org/0000-0001-5024-8272**):** Data curation, Writing – reviewing and editing.

Christopher Vernazza (https://orcid.org/0000-0002-6927-2974): Writing – reviewing and editing.

Mike Bradburn: Conceptualisation, Funding acquisition, Methodology.

Jaimie Buckley: Conceptualisation, Funding acquisition, Methodology, PPI.

Julie Child-Cavill: Conceptualisation, Funding acquisition, Methodology, PPI.

Ayman Eissa: Conceptualisation, Funding acquisition, Methodology.

Graham Wilson: Conceptualisation, Funding acquisition, Methodology.

Oscar Bortolami: Conceptualisation, Funding acquisition.

Jamie Hall: Data curation, Investigation.

Anna Thomason: Data curation, Investigation.

Katie Mellor: Data curation, Investigation.

Simon Waterhouse: Data curation.

John Rouse: PPI.

Acknowledgements

We gratefully acknowledge the hard work, support and advice from the following: the MAGIC study teams, research nurses; research support staff and clinical teams within the participating NHS trusts for participant screening, delivering the intervention, data collection and patient follow-up. We also offer thanks to Alessia Dunn and Erica Wallis (research managers as Research Sponsor) for support of the trial. We specially thank the 110 patients who took part in the study, as well as their families for supporting their choice and also contributing.

We thank and acknowledge advice and oversight from members of the Trial Steering Committee: Professor Anne Schilder (Chair), Dr Paul Brady, Dr Justine Heard, Dr Nia Goulden and Mr John Rouse (PPI) and Data Monitoring and Ethics Committee: Professor Ivor Chestnutt (Chair), Mr Lee Middleton, Dr James Armstrong and Dr Glyn Williams. We also thank Chris Evans and team for the use of the Little Journey mYPAS-SF training platform. We also acknowledge trial support and contributions from Mike Bradburn, Jamie Buckley, Julie Child-Cavill, Munya Dimairo, Enass Duro, Ayman Eissa, Katie Mellor, Sian Rolfe, Anna Thomason, Graham Wilson, Simon Waterhouse and Julian Yates.

Patient data statement

This work uses data provided by patients and collected by the NHS as part of their care and support. Using patient data is vital to improve health and care for everyone. There is huge potential to make better use of information from people's patient records, to understand more about disease, develop new treatments, monitor safety and plan NHS services. Patient data should be kept safe and secure, to protect everyone's privacy, and it is important that there are safeguards to make sure that they are stored and used responsibly. Everyone should be able to find out about how patient data are used. #datasaveslives You can find out more about the background to this citation here: https://understandingpatientdata.org.uk/data-citation.

Data-sharing statement

All data requests should be submitted to the corresponding author for consideration. Access to anonymised data may be granted following review.

Ethics statement

The trial was approved by North West – Liverpool Central Research Ethics Committee (18/NW/0758) on 16 January 2019.

Information governance statement

The NIHR is committed to handling all personal information in line with the UK Data Protection Act (2018) and the General Data Protection Regulation (EU GDPR) 2016/679. Under Data Protection legislation, the University of Sheffield is the Data Processor; Sheffield Teaching Hospitals NHS Foundation Trust is the Data Controller, and we process personal data in accordance with their instructions. You can find out more about how we handle personal data, including how to exercise your individual rights and the contact details for University of Sheffield's Data Protection Officer (www.sheffieldclinicalresearch.org/).

This synopsis should be referenced as follows:

Deery C, Bolt R, Papaioannou D, Wilson M, Hyslop M, Herbert E, et al. Melatonin versus midazolam in the premedication of anxious children attending for elective surgery under general anaesthesia: the MAGIC non-inferiority RCT. Health Technol Assess 2025;29(29). https://doi.org/10.3310/CWKF1987

Disclosure of interests

Full disclosure of interests: Completed ICMJE forms for all authors, including all related interests, are available in the toolkit on the NIHR Journals Library report publication page at https://doi.org/10.3310/CWKF1987.

Primary conflicts of interest: Christopher Deery – Grants from: IADR ERD, NIHR Doctorial fellowship, FDS RCSEng British Society of Paediatric Dentistry (FDS-BSPD) Pump-Priming Grants Scheme, Ministry of Education of Brazil, Colgate Robin Davies/ Oral Dental Research Trust DCP – Research Awards, NIHR129230 Pulpotomy for Irreversible Pulpitis – PIP Trial, NIHR 17/127/07 Selective Caries Removal in Permanent Teeth (SCRIPT), PhD Studentship, Saudi Arabian Government, Consulting Fees from: Johnson and Johnson Listerine and National Advisory Panel/Panel Member Oman Academic Accreditation Authority Institutional Accreditation, Standards Assessment.

Matthew Wilson – Member of: Data Monitoring and Ethics Committee membership RESULT Hip Trial.

Laura Flight – Grants from: the NIHR CTU Support and NIHR Doctoral Fellowship Schemes.

Fiona Gilchrist - HTA PCCPI Panel 2018-2019 HTA Prioritisation Committee C (Mental health, women and children's health) 2018-9.

Helen Rodd – Chair of the Research Committee, Faculty of Dental Surgery, Royal College of Surgeons, England.

Christopher Vernazza – Grants from: UKRI Horizon Europe Guarantee Funding (PRUDENT), Borrow Foundation (FLOWAVE), NIHR HTA (CALM), QuantumDx.

Department of Health and Social Care disclaimer

This publication presents independent research commissioned by the NIHR. The views and opinions expressed by authors in this publication are those of the authors and do not necessarily reflect those of the NHS, the NIHR, MRC, NIHR Coordinating Centre, the Health Technology Assessment programme or the Department of Health and Social Care.

This synopsis was published based on current knowledge at the time and date of publication. NIHR is committed to being inclusive and will continually monitor best practice and guidance in relation to terminology and language to ensure that we remain relevant to our stakeholders.

Trial registration

16

This trial is registered as ISRCTN18296119.

Funding

This synopsis presents independent research funded by the National Institute for Health and Care Research (NIHR) Health Technology Assessment programme as award number 16/80/08.

Award publications

This synopsis provided an overview of the research award The MAGIC trial (Melatonin for Anxiety prior to General anaesthesia In Children): A Multicentre, Parallel Randomised Controlled Trial of Melatonin Versus Midazolam in the Premedication of Anxious Children Attending for Elective Dental and ENT Surgery Under General Anaesthesia. Other articles published as part of this thread are:

Bolt R, Hyslop MC, Herbert E, Papaioannou DE, Totton N, Wilson MJ, *et al.* The MAGIC trial: a pragmatic, multicentre, parallel, noninferiority, randomised trial of melatonin versus midazolam in the premedication of anxious children attending for elective surgery under general anaesthesia, *Br J Anaesth* 2024;**132**:76–85. https://doi.org/10.1016/j.bja.2023.10.011

Kettle J, Bolt R, Deery C, Papaioannou D, Rodd H, Hyslop MC, *et al.* Acceptability of midazolam and melatonin as premedications for anxious children undergoing general anaesthetic: a qualitative interview study with children, caregivers and health professionals participating in the MAGIC randomised controlled trial. *Trials* 2024;25:813. https://doi.org/10.1186/s13063-024-08611-x

Kettle J, Deery C, Bolt R, Papaioannou D, Marshman Z. Stakeholder perspectives on barriers and enablers to recruiting anxious children undergoing day surgery under general anaesthetic: a qualitative internal pilot study of the MAGIC randomised controlled trial. *Trials* 2021;**22**:458. https://doi.org/10.1186/s13063-021-05425-z

Hyslop MC, Papaioannou D, Bolt R, Wilson M, Bradburn M, Clarkson J, *et al.* Barriers and enablers to recruiting participants within perioperative and anaesthetic settings: lessons learned from the MAGIC trial. *BJA Online* 2025;**13**:100375. https:// doi.org/10.1016/j.bjao.2024.100375

For more information about this research please view the award page (https://fundingawards.nihr.ac.uk/award/16/80/08).

Additional outputs

Mellor K, Papaioannou D, Thomason A, Bolt R, Evans C, Wilson M, Deery C. Melatonin for pre-medication in children: a systematic review. *BMC Pediatr* 2022;**22**:107. https://doi.org/10.1186/ s12887-022-03149-w Deery C, Bolt R, Papaioannou D, Totton N, Herbert E, Hyslop M, et al. The MAGIC Trial (Melatonin for Anxiety prior to General anaesthesia In Children): A Multicentre, Parallel Randomised Controlled Trial of Melatonin versus Midazolam in the Premedication of Anxious Children Attending for Elective Surgery Under General Anaesthesia. The University of Sheffield. Workflow; 2023. https://doi.org/10.15131/shef.data.22220884.v1

Herbert E, Totton N, Deery C, Bolt R, Hyslop M, Bradburn M, *et al.* MAGIC *Trial – Statistical Analysis Plan.* The University of Sheffield. Workflow; 2023. https://doi.org/10.15131/shef.data.23180222

Herbert E, Papaioannou D, Loban A, Totton N, Hyslop M, Bolt R, Deery C. Personalised versus standard text message prompts for increasing trial participant response to telephone follow up: an embedded randomised controlled retention trial. *Trials* 2024;**25**:108. https://doi.org/10.1186/s13063-024-07916-1

About this synopsis

The contractual start date for this research was in December 2017. This article began editorial review in November 2023 and was accepted for publication in October 2024. The authors have been wholly responsible for all data collection, analysis and interpretation, and for writing up their work. The Health Technology Assessment editors and publisher have tried to ensure the accuracy of the authors' article and would like to thank the reviewers for their constructive comments on the draft document. However, they do not accept liability for damages or losses arising from material published in this article.

Copyright

Copyright © 2025 Deery *et al.* This work was produced by Deery *et al.* under the terms of a commissioning contract issued by the Secretary of State for Health and Social Care. This is an Open Access publication distributed under the terms of the Creative Commons Attribution CC BY 4.0 licence, which permits unrestricted use, distribution, reproduction and adaptation in any medium and for any purpose provided that it is properly attributed. See: https://creativecommons.org/licenses/by/4.0/. For attribution the title, original author(s), the publication source - NIHR Journals Library, and the DOI of the publication must be cited.

Published by the NIHR Journals Library (www.journalslibrary. nihr.ac.uk), produced by Newgen Digitalworks Pvt Ltd, Chennai, India (www.newgen.co).

List of abbreviations

AE	adverse event
ENT	ear, nose and throat

EU	European Union
FPS-R	Revised Faces Pain Scale (observer and participant reported)
GA	general anaesthetic
HTA	Health Technology Assessment programme
IMP	investigational medicinal product
ITT	intention to treat
MAGIC	Melatonin for Anxiety prior to General anaesthesia In Children
MYPAS-SF	Modified Yale Preoperative Anxiety Scale Short Form
NIHR	National Institute for Health and Care Research
PAED	Paediatric Anaesthesia Emergence Delirium scale index
PHBQ-AS	Post Hospitalisation Behaviour Questionnaire for Ambulatory Surgery
PI	principal investigator
PP	per-protocol
PPI	patient and public involvement
QALY	quality-adjusted life-year
RCT	randomised controlled trial
SAE	serious adverse event
SWAT	study within a trial
TMG	Trial Management Group
TSC	Trial Steering Committee
VSRS	Vancouver Sedation Recovery Scale

References

- Deery C, Bolt R, Papaioannou D, Totton N, Herbert E, Hyslop M, et al. The MAGIC trial (Melatonin for Anxiety prior to General anaesthesia In Children): A Multicentre, Parallel Randomised Controlled Trial of Melatonin versus Midazolam in the Premedication of Anxious Children Attending for Elective Surgery under General Anaesthesia. The University of Sheffield. Workflow; 2023. https:// doi.org/10.15131/shef.data.22220884.v1
- Bolt R, Hyslop M, Herbert E, Papaioannou D, Totton N, Deery C, et al. The MAGIC trial – a multicentre, parallel, non-inferiority, randomised controlled trial of

This synopsis should be referenced as follows:

Deery C, Bolt R, Papaioannou D, Wilson M, Hyslop M, Herbert E, et al. Melatonin versus midazolam in the premedication of anxious children attending for elective surgery under general anaesthesia: the MAGIC non-inferiority RCT. Health Technol Assess 2025;29(29). https://doi.org/10.3310/CWKF1987

melatonin versus midazolam in the premedication of anxious children attending for elective surgery under general anaesthesia. *Br J Anaesth* 2024;**132**:76–85. https://doi.org/10.1016/j.bja.2023.10.011

- Sury MR, Arumainathan R, Belhaj AM, MacG Palmer JH, Cook TM, Pandit JJ. The state of UK pediatric anesthesia: a survey of National Health Service activity. *Paediatr Anaesth* 2015;25:1085–92. https://doi. org/10.1111/pan.12753
- NHS Digital. Hospital Admitted Patient Care Activity, 2015–16: Summary Report. URL: http://content.digital. nhs.uk/catalogue/PUB22378/hosp-epis-stat-admisumm-rep-2015-16-rep.pdf (accessed 9 November 2023).
- Chorney JM, Torrey C, Blount R, McLaren CE, Chen WP, Kain ZN. Healthcare provider and parent behavior and children's coping and distress at anesthesia induction. *Anesthesiology* 2009;**111**:1290–6. https:// doi.org/10.1097/ALN.0b013e3181c14be5
- Kocherov S, Hen Y, Jaworowski S, Ostrovsky I, Eidelman Al, Gozal Y, Chertin B. Medical clowns reduce pre-operative anxiety, post-operative pain and medical costs in children undergoing outpatient penile surgery: a randomised controlled trial. *J Paediatr Child Health* 2016;52:877–81. https://doi.org/10.1111/ jpc.13242
- Astuto M, Rosano G, Rizzo G, Disma N, Raciti L, Sciuto O. Preoperative parental information and parents' presence at induction of anaesthesia. *Minerva Anestesiol* 2006;**72**:461–5.
- Arai YC, Kandatsu N, Ito H, Sato Y, Satake Y, Mizutani M, et al. Induction and emergence behavior of children undergoing general anesthesia correlates with maternal salivary amylase activity before the anesthesia. *Acta Anaesthesiol Scand* 2008;52:285–8. https://doi. org/10.1111/j.1399-6576.2007.01527.x
- Gooden R, Tennant I, James B, Augier R, Crawford-Sykes A, Ehikhametalor K, et al. Incidência de delírio ao despertar e fatores de risco após o uso de sevoflurano em pacientes pediátricos para cirurgia ambulatorial, Kingston, Jamaica [The incidence of emergence delirium and risk factors following sevoflurane use in pediatric patients for day case surgery, Kingston, Jamaica]. *Rev Bras Anestesiol* 2014;64:413–8. https:// doi.org/10.1016/j.bjan.2013.09.011
- Kain ZN, Mayes LC, Caldwell-Andrews AA, Karas DE, McClain BC. Preoperative anxiety, postoperative pain, and behavioral recovery in young children undergoing surgery. *Pediatrics* 2006;**118**:651–8. https://doi. org/10.1542/peds.2005-2920
- 11. McCluskey A, Meakin GH. Oral administration of midazolam as a premedicant for paediatric day-case

anaesthesia. *Anaesthesia* 1994;**49**:782–5. https://doi. org/10.1111/j.1365-2044.1994.tb04451.x

- 12. Cray SH, Dixon JL, Heard CM, Selsby DS. Oral midazolam premedication for paediatric day case patients. *Paediatr Anaesth* 1996;**6**:265–70. https://doi. org/10.1111/j.1460-9592.1996.tb00448.x
- 13. Viitanen H, Annila P, Viitanen M, Tarkkila P. Premedication with midazolam delays recovery after ambulatory sevoflurane anesthesia in children. *Anesth Analg* 1999;**89**:75–9. https://doi. org/10.1097/00000539-199907000-00014
- 14. Sievers TD, Yee JD, Foley ME, Blanding PJ, Berde CB. Midazolam for conscious sedation during pediatric oncology procedures: safety and recovery parameters. *Pediatrics* 1991;**88**:1172–9.
- 15. Maeda S, Tomoyasu Y, Higuchi H, Mori T, Egusa M, Miyawaki T. Midazolam is associated with delay in recovery and agitation after ambulatory general anesthesia for dental treatment in patients with disabilities: a retrospective cohort study. *J Oral Maxillofac Surg* 2012;70:1315-20. https://doi.org/10.1016/j. joms.2012.01.004
- 16. Pokharel K, Tripathi M, Gupta PK, Bhattarai B, Khatiwada S, Subedi A. Premedication with oral alprazolam and melatonin combination: a comparison with either alone – a randomized controlled factorial trial. *Biomed Res Int* 2014;2014:356964. https://doi. org/10.1155/2014/356964
- Khezri MB, Merate H. The effects of melatonin on anxiety and pain scores of patients, intraocular pressure, and operating conditions during cataract surgery under topical anesthesia. *Indian J Ophthalmol* 2013;61:319–24. https://doi.org/10.4103/0301-4738.99637
- 18. Ismail SA, Mowafi HA. Melatonin provides anxiolysis, enhances analgesia, decreases intraocular pressure, and promotes better operating conditions during cataract surgery under topical anesthesia. *Anesth Analg* 2009;**108**:1146–51. https://doi.org/10.1213/ ane.0b013e3181907ebe
- 19. Mowafi HA, Ismail SA. Melatonin improves tourniquet tolerance and enhances postoperative analgesia in patients receiving intravenous regional anesthesia. *Anesth Analg* 2008;**107**:1422–6. https:// doi.org/10.1213/ane.0b013e318181f689
- Caumo W, Torres F, Moreira NL Jr, Auzani JA, Monteiro CA, Londero G, *et al.* The clinical impact of preoperative melatonin on postoperative outcomes in patients undergoing abdominal hysterectomy. *Anesth Analg* 2007;**105**:1263–71, table of contents. https://doi. org/10.1213/01.ane.0000282834.78456.90

- 21. Naguib M, Samarkandi AH. Premedication with melatonin: a double-blind, placebo-controlled comparison with midazolam. Br J Anaesth 1999;82:875-80. https:// doi.org/10.1093/bja/82.6.875
- 22. Naguib M, Samarkandi AH. The comparative dose-response effects of melatonin and midazolam for premedication of adult patients: double-blinded, placebo-controlled study. а 2000;91:473-9. Anesth Analg https://doi. org/10.1097/00000539-200008000-00046
- 23. Andersen LP, Gögenur I, Rosenberg J, Reiter RJ. The safety of melatonin in humans. Clin Drug Investig 2016;36:169-75. https://doi.org/10.1007/ s40261-015-0368-5
- 24. National Institute for Health and Care Excellence. Melatonin (BNF). URL: https://bnf.nice.org.uk/drugs/ melatonin/ (accessed 9 November 2023).
- 25. Mellor K, Papaioannou D, Thomason A, Bolt R, Evans C, Wilson M, Deery C. Melatonin for pre-medication in children: a systematic review. BMC Pediatr 2022;22:107. https://doi.org/10.1186/ s12887-022-03149-w
- 26. Jenkins BN, Kain ZN, Kaplan SH, Stevenson RS, Mayes LC, Guadarrama J, Fortier MA. Revisiting a measure of child postoperative recovery: development of the post hospitalization behavior questionnaire for ambulatory surgery. Paediatr Anaesth 2015;25:738-45. https:// doi.org/10.1111/pan.12678
- 27. Kain ZN, Mayes LC, Cicchetti DV, Caramico LA, Spieker M, Nygren MM, Rimar S. Measurement tool for preoperative anxiety in young children: the Yale Preoperative Anxiety Scale. Child Neuropsychol 1995;1:203-10. https://doi.org/10.1080/09297049508400225
- 28. Jenkins BN, Fortier MA, Kaplan SH, Mayes LC, Kain ZN. Development of a short version of the modified Yale Preoperative Anxiety Scale. Anesth Analg 2014;119:643-50. https://doi.org/10.1213/ ANE.00000000000350
- 29. Sikich N, Lerman J. Development and psychometric evaluation of the pediatric anesthesia emergence delirium scale. Anesthesiology 2004;100:1138-45. https://doi.org/10.1097/00000542-200405000-00015
- 30. Macnab AJ, Levine M, Glick N, Susak L, Baker-Brown G. A research tool for measurement of recovery from sedation: the Vancouver Sedative Recovery Scale. J Pediatr Surg 1991;26:1263-7. https://doi. org/10.1016/0022-3468(91)90594-j
- 31. McGrath PA. Seifert CE, Speechley KN. Booth JC, Stitt L, Gibson MC. A new analogue scale for assessing children's pain: an initial

validation study. Pain 1996;64:435-43. https://doi. org/10.1016/0304-3959(95)00171-9

- 32. Herbert E, Totton N, Deery C, Bolt R, Hyslop M, Bradburn M, et al. MAGIC Trial - Statistical Analysis Plan. The University of Sheffield. Workflow; 2023. https://doi.org/10.15131/shef.data.23180222
- 33. Julian LJ. Measures of anxiety: State-Trait Anxiety Inventory (STAI), Beck Anxiety Inventory (BAI), and Hospital Anxiety and Depression Scale-Anxiety (HADS-A). Arthritis Care Res (Hoboken) 2011;63:S467-72. https://doi.org/10.1002/acr.20561
- 34. Ratcliffe J, Huynh E, Chen G, Stevens K, Swait J, Brazier J, et al. Valuing the Child Health Utility 9D: using profile case best worst scaling methods to develop a new adolescent specific scoring algorithm. Soc Sci 2016;157:48-59. https://doi.org/10.1016/j. Med socscimed.2016.03.042
- 35. Kettle J, Deery C, Bolt R, Papaioannou D, Marshman Z. Stakeholder perspectives on barriers and enablers to recruiting anxious children undergoing day surgery under general anaesthetic: a qualitative internal pilot study of the MAGIC randomised controlled trial. Trials 2021;22:458. https://doi.org/10.1186/ s13063-021-05425-z
- 36. Young T, Bolt R, Flight L, Herbert E, Hyslop M, Deery C, et al. The MAGIC Trial (Melatonin for Anxiety prior to General anaesthesia In Children) - Health Economics Report. Personal Communication.
- 37. Drummond M, McGuire A. Economic Evaluation in Health Care: Merging Theory with Practice. New York: Oxford University Press; 2001.
- 38. Kurdi MS, Muthukalai SP. A comparison of the effect of two doses of oral melatonin with oral midazolam and placebo on pre-operative anxiety, cognition and psychomotor function in children: a randomised double-blind study. Indian J Anaesth 2016;60:744-50. https://doi.org/10.4103/0019-5049.191688
- 39. Samarkandi A, Naguib M, Riad W, Thalaj A, Alotibi W, Aldammas F, Albassam A. Melatonin vs. midazolam premedication in children: a double-blind, placebocontrolled study. Eur J Anaesthesiol 2005;22:189-96. https://doi.org/10.1017/s0265021505000335
- 40. Marseglia L, D'Angelo G, Manti S, Aversa S, Arrigo T, Reiter RJ, Gitto E. Analgesic, anxiolytic and anaesthetic effects of melatonin: new potential uses in pediatrics. Int J Mol Sci 2015;16:1209-20. https://doi. org/10.3390/ijms16011209
- 41. Thoma A, Farrokhyar F, McKnight L, Bhandari M. Practical tips for surgical research: how to optimize patient recruitment. Can J Surg 2010;53:205-10.

This synopsis should be referenced as follows: Deery C, Bolt R, Papaioannou D, Wilson M, Hyslop M, Herbert E, et al. Melatonin versus midazolam in the premedication of anxious children attending for elective surgery under general anaesthesia: the MAGIC non-inferiority RCT. Health Technol Assess 2025;29(29). https://doi.org/10.3310/CWKF1987

- 42. Lasagna L. Problems in publication of clinical trial methodology. *Clin Pharmacol Ther* 1979;**25**:751–3. https://doi.org/10.1002/cpt1979255part2751
- 43. Team V, Bugeja L, Weller CD. Barriers and facilitators to participant recruitment to randomised controlled trials: a qualitative perspective. *Int Wound J* 2018;15:929-42. https://doi.org/10.1111/iwj. 12950
- 44. Yap CYL, Hsueh YA, Knott JC, Taylor DM, Chan EW, Kong DCM. Economic evaluation of midazolamdroperidol combination, versus droperidol or olanzapine for the management of acute agitation in the emergency department: a within-trial analysis. *PharmacoEconomics Open* 2018;2:141–51. https://doi. org/10.1007/s41669-017-0047-y
- 45. Wolf A, McKay A, Spowart C, Granville H, Boland A, Petrou S, et al. Prospective multicentre randomised, double-blind, equivalence study comparing clonidine and midazolam as intravenous sedative agents in critically ill children: the SLEEPS (Safety profiLe, Efficacy and Equivalence in Paediatric intensive care Sedation) study. *Health Technol Assess* 2014;**18**:1–212. https:// doi.org/10.3310/hta18710
- 46. Hohl CM, Nosyk B, Sadatsafavi M, Anis AH. A costeffectiveness analysis of propofol versus midazolam for procedural sedation in the emergency department. Acad Emerg Med 2008;15:32–9. [Erratum published in Acad Emerg Med 2008;15:304.]. https:// doi.org/10.1111/j.1553-2712.2007.00023.x
- 47. National Clinical Guideline Centre (UK). Sedation in Children and Young People: Sedation for Diagnostic and Therapeutic Procedures in Children and Young People (Appendix F). London: Royal College of Physicians (UK); 2010. (NICE Clinical Guidelines, No. 112.). URL: www. nice.org.uk/guidance/cg112/evidence/full-guideline-136287325 (accessed 9 November 2023).
- 48. Wang L, Huang L, Zhang T, Peng W. Comparison of intranasal dexmedetomidine and oral midazolam for premedication in pediatric dental patients under general anesthesia: a randomised clinical trial.

Biomed Res Int 2020;2020:5142913. https://doi. org/10.1155/2020/5142913

- 49. Chorney JM, Kain ZN. Behavioral analysis of children's response to induction of anesthesia. *Anesth Analg* 2009;**109**:1434–40. https://doi.org/10.1213/ ane.0b013e3181b412cf
- 50. van Dorn A. COVID-19 and readjusting clinical trials. Lancet 2020;**396**:523-4. https://doi.org/10.1016/ S0140-6736(20)31787-6
- Jackson EL, Feldschreiber P, Breckenridge A. Regulatory consequences of 'brexit' for the development of medicinal products. *Clin Pharmacol Ther* 2017;102:183–4. https://doi.org/10.1002/cpt.706
- 52. Yang CQ, Yu KH, Huang RR, Qu SS, Zhang JM, Li YL. Comparison of different sedatives in children before general anaesthesia for selective surgery: a network meta-analysis. J Clin Pharm Ther 2022;47:1495-505. https://doi.org/10.1111/jcpt.13763
- 53. Levine RS. Childhood caries and hospital admissions in England: a reflection on preventive strategies. *Br Dent J* 2021;**230**:611–6. https://doi.org/10.1038/ s41415-021-2945-8
- 54. Hyslop MC, Papaioannou D, Bolt R, Wilson M, Bradburn M, Clarkson J, *et al.* Barriers and enablers to recruiting participants within perioperative and anaesthetic settings: lessons learned from the MAGIC trial. *BJA Online* 2025; in press.
- 55. Treweek S, Bevan S, Bower P, Campbell M, Christie J, Clarke M, *et al.* Trial forge guidance 1: what is a Study Within A Trial (SWAT)? *Trials* 2018;**19**:139. https://doi. org/10.1186/s13063-018-2535-5
- 56. Herbert E, Papaioannou D, Loban A, Totton N, Hyslop M, Bolt R, Deery C. Personalised versus standard text message prompts for increasing trial participant response to telephone follow-up: an embedded randomised controlled retention trial. *Trials* 2024;25:108. https://doi.org/10.1186/ s13063-024-07916-1

Appendix 1

TABLE 2 Summary of changes to the MAGIC trial protocol

Changes to protocol	Date	Approved by
Protocol version 2.0 (not implemented): Updated in response to REC request to remove the £10 vouchers for qualitative study interviewees	19 December 2018	N/A
Protocol version 3.0 (approved version on trial opening): Updates included change from PHBQ to PHBQ-AS; removal of post box test; update to non-permitted medication; change to allow verbal assent for all children; change to allow for nurse prescribers; change to allow for postal return CHU9D questionnaires at follow-up; change of timing for postoperative assessments to every 15 minutes from every 10 minutes; removal of out of hours unblinding system; replacement of 'until stage 2 recovery completion'	27 March 2019	North West – Liverpool Central Research Ethics Committee Health Research Authority MHRA
Protocol version 4.0: Updates included change to expand inclusion criteria to allow 3- and 4-year-old children to be recruited; change to expand inclusion criteria to include more surgical specialties: gastroenterology, radiology, plastic, orthopaedic, urology or other general surgery; change to clarify assent requirement from children. Children who neither provide assent nor decline the trial (due to high anxiety) can be enrolled based on caregiver consent and PI decision. Children who verbally decline to participate must not be included; change for those sites who do not have dedicated preoperative clinics, to allow the team to send study information prior to the day of surgery. This will be based on PI decision of the suitability of the participant to receive this information; change to randomisation system from stratification to minimisation; change to allow children the option of reviewing the information sheet or the video and not a requirement to undertake both; clarification of secondary safety and efficacy objectives and outcomes and addition of CHU9D proxy questionnaire for children aged 3-4	7 May 2020	North West – Liverpool Central Research Ethics Committee Health Research Authority MHRA
Protocol version 4.1: Updates included change to clarify assent from highly anxious children not mandatory and can be based on caregiver and PI (or delegated individual) decision alone; change to clarify baseline assessments can be undertaken after randomisation; change to allow remote consent and interviews for the qualitative substudy in light of the 2020 COVID-19 pandemic	28 September 2020	Non-substantial – no approvals required

MHRA, Medicines and Healthcare products Regulatory Agency.

Note

Reproduced with permission from Bolt et al.² This is an Open Access article distributed under the terms of the Creative Commons Attribution Licence, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited. This table includes minor additions and formatting changes to the original text.

Appendix 2

			Midaz	olam	Melat	onin	 Adjusted mean difference (95% CI)
Score	Analysis population	Time point	n	Mean (SD)	n	Mean (SD)	
VSRS	ITT	15	15	12.73 (5.19)	20	14.35 (4.79)	0.122 (-1.567 to 1.811)
		30	32	14.28 (4.89)	26	17.27 (5.77)	0.122 (-1.567 to 1.811)
		45	32	17.28 (4.10)	31	18.35 (5.08)	0.122 (-1.567 to 1.811)
		60	29	18.90 (4.39)	28	19.04 (4.86)	0.122 (-1.567 to 1.811)
		75	25	20.80 (2.87)	25	19.72 (4.27)	0.122 (-1.567 to 1.811)
		90	22	20.68 (3.86)	21	19.33 (3.97)	0.122 (-1.567 to 1.811)
		105	19	20.42 (3.99)	18	18.94 (4.12)	0.122 (-1.567 to 1.811)
		120	14	21.14 (2.41)	13	19.69 (4.15)	0.122 (-1.567 to 1.811)
	PP	15	15	12.73 (5.19)	18	14.50 (5.03)	0.413 (-1.331 to 2.156)
		30	31	14.29 (4.97)	22	18.09 (5.58)	0.413 (-1.331 to 2.156)
		45	31	17.13 (4.07)	27	18.89 (4.70)	0.413 (-1.331 to 2.156)
		60	28	18.79 (4.43)	23	19.91 (4.18)	0.413 (-1.331 to 2.156)
		75	24	20.75 (2.92)	21	19.95 (4.52)	0.413 (-1.331 to 2.156)
		90	21	20.62 (3.94)	18	19.67 (4.03)	0.413 (-1.331 to 2.156)
		105	19	20.42 (3.99)	15	18.93 (4.43)	0.413 (-1.331 to 2.156)
		120	14	21.14 (2.41)	10	19.00 (4.55)	0.413 (-1.331 to 2.156)
Co-operation Score	ITT	15	38	0.92 (2.02)	38	1.39 (2.57)	-0.064 (-0.843 to 0.716)
		30	43	3.02 (2.81)	35	3.26 (3.25)	-0.064 (-0.843 to 0.716)
		45	37	4.16 (2.69)	36	4.64 (2.95)	-0.064 (-0.843 to 0.716)
		60	30	5.37 (2.40)	34	5.06 (2.80)	-0.064 (-0.843 to 0.716)
		75	31	5.90 (1.99)	31	5.32 (2.76)	-0.064 (-0.843 to 0.716)
		90	25	6.16 (1.60)	24	6.00 (2.13)	-0.064 (-0.843 to 0.716)
		105	20	6.40 (1.39)	21	6.24 (2.12)	-0.064 (-0.843 to 0.716)

			Midaz	olam	Melate	onin	
Score	Analysis population	Time point	n	Mean (SD)	n	Mean (SD)	Adjusted mean difference (95% CI)
		120	15	6.27 (1.44)	17	6.53 (1.70)	-0.064 (-0.843 to 0.716)
	PP	15	37	0.95 (2.04)	32	1.25 (2.44)	0.047 (-0.764 to 0.858)
		30	42	2.98 (2.82)	32	3.44 (3.29)	0.047 (-0.764 to 0.858)
		45	36	4.19 (2.72)	32	4.69 (3.05)	0.047 (-0.764 to 0.858)
		60	29	5.31 (2.42)	29	5.24 (2.82)	0.047 (-0.764 to 0.858)
		75	30	5.87 (2.01)	26	5.88 (2.41)	0.047 (-0.764 to 0.858)
		90	24	6.12 (1.62)	20	6.50 (1.61)	0.047 (-0.764 to 0.858)
		105	20	6.40 (1.39)	18	6.22 (2.26)	0.047 (-0.764 to 0.858)
		120	15	6.27 (1.44)	14	6.50 (1.87)	0.047 (-0.764 to 0.858)
FPS-R (child reported)	ITT	15	4	5.00 (4.76)	5	3.40 (3.97)	0.170 (-0.388 to 0.729)
		30	16	2.31 (3.46)	14	2.43 (3.25)	0.170 (-0.388 to 0.729)
		45	21	2.86 (3.44)	22	1.64 (2.11)	0.170 (-0.388 to 0.729)
		60	22	3.23 (3.54)	24	2.54 (3.53)	0.170 (-0.388 to 0.729)
		75	23	2.22 (2.59)	22	2.23 (3.12)	0.170 (-0.388 to 0.729)
		90	19	1.53 (2.32)	20	2.30 (3.13)	0.170 (-0.388 to 0.729)
		105	18	2.44 (3.11)	17	2.00 (2.45)	0.170 (-0.388 to 0.729)
		120	12	0.67 (1.30)	14	1.71 (2.20)	0.170 (-0.388 to 0.729)
	PP	15	4	5.00 (4.76)	4	1.75 (1.71)	0.126 (-0.419 to 0.671)
		30	15	2.47 (3.52)	14	2.43 (3.25)	0.126 (-0.419 to 0.671)
		45	20	3.00 (3.46)	21	1.71 (2.12)	0.126 (-0.419 to 0.671)
		60	21	3.29 (3.62)	22	2.77 (3.60)	0.126 (-0.419 to 0.671)
		75	22	2.23 (2.65)	21	2.33 (3.15)	0.126 (-0.419 to 0.671)
		90	18	1.50 (2.38)	17	2.71 (3.24)	0.126 (-0.419 to 0.671)
							continued

This synopsis should be referenced as follows: Deery C, Bolt R, Papaioannou D, Wilson M, Hyslop M, Herbert E, et al. Melatonin versus midazolam in the premedication of anxious children attending for elective surgery under general anaesthesia: the MAGIC non-inferiority RCT. Health Technol Assess 2025;29(29). https://doi.org/10.3310/CWKF1987

			Midaz	olam	Melat	onin		
Score	Analysis population	Time point	n	Mean (SD)	n	Mean (SD)	Adjusted mean difference (95% CI)	
		105	18	2.44 (3.11)	15	2.27 (2.49)	0.126 (-0.419 to 0.671)	
		120	12	0.67 (1.30)	11	2.00 (2.37)	0.126 (-0.419 to 0.671)	
FPS-R (observer reported)	ITT	15	27	2.44 (3.20)	29	1.14 (1.73)	-0.046 (-0.585 to 0.493)	
		30	37	2.38 (2.75)	30	1.83 (2.49)	-0.046 (-0.585 to 0.493)	
		45	34	2.38 (2.93)	36	2.28 (2.87)	-0.046 (-0.585 to 0.493)	
		60	31	2.32 (2.37)	32	1.91 (3.03)	-0.046 (-0.585 to 0.493)	
		75	31	1.61 (2.09)	31	1.65 (2.80)	-0.046 (-0.585 to 0.493)	
		90	26	1.73 (2.25)	25	1.00 (1.91)	-0.046 (-0.585 to 0.493)	
		105	20	0.85 (1.35)	21	1.33 (2.24)	-0.046 (-0.585 to 0.493)	
		120	15	1.13 (1.81)	16	0.94 (1.77)	-0.046 (-0.585 to 0.493)	
	PP	15	27	2.44 (3.20)	24	1.12 (1.75)	0.013 (-0.610 to 0.636)	
		30	36	2.44 (2.76)	26	2.04 (2.60)	0.013 (-0.610 to 0.636)	
		45	33	2.45 (2.95)	31	2.58 (2.98)	0.013 (-0.610 to 0.636)	
		60	30	2.33 (2.41)	27	2.26 (3.18)	0.013 (-0.610 to 0.636)	
		75	30	1.60 (2.13)	26	1.96 (2.96)	0.013 (-0.610 to 0.636)	
		90	25	1.72 (2.30)	21	1.19 (2.04)	0.013 (-0.610 to 0.636)	
		105	20	0.85 (1.35)	18	1.56 (2.36)	0.013 (-0.610 to 0.636)	
		120	15	1.13 (1.81)	13	1.15 (1.91)	0.013 (-0.610 to 0.636)	
PAED	ITT	15	40	11.97 (2.36)	39	11.31 (3.74)	-0.699 (-2.039 to 0.641)	
		30	43	10.05 (4.38)	38	8.32 (5.39)	-0.699 (-2.039 to 0.641)	
		45	38	8.08 (4.24)	38	6.45 (5.05)	-0.699 (-2.039 to 0.641)	
		60	30	5.53 (3.95)	34	5.12 (5.08)	-0.699 (-2.039 to 0.641)	
		75	31	4.29 (4.04)	32	4.41 (4.74)	-0.699 (-2.039 to 0.641)	
		90	25	2.96 (3.86)	25	3.16 (4.22)	-0.699 (-2.039 to 0.641)	
		105	20	1.60 (2.52)	20	3.45 (3.94)	-0.699 (-2.039 to 0.641)	

		Mida		Midazolam Me		onin	
Score	Analysis population	Time point	n	Mean (SD)	n	Mean (SD)	Adjusted mean difference (95% Cl)
		120	15	1.40 (3.09)	17	2.94 (3.38)	-0.699 (-2.039 to 0.641)
	PP	15	39	11.97 (2.39)	33	11.39 (3.77)	-0.856 (-2.167 to 0.456)
		30	42	10.05 (4.43)	33	7.97 (5.46)	-0.856 (-2.167 to 0.456)
		45	37	8.03 (4.28)	33	6.18 (5.20)	-0.856 (-2.167 to 0.456)
		60	29	5.59 (4.01)	29	4.83 (5.20)	-0.856 (-2.167 to 0.456)
		75	30	4.30 (4.11)	27	3.93 (4.71)	-0.856 (-2.167 to 0.456)
		90	24	3.00 (3.93)	21	2.57 (3.78)	-0.856 (-2.167 to 0.456)
		105	20	1.60 (2.52)	17	3.41 (4.21)	-0.856 (-2.167 to 0.456)
		120	15	1.40 (3.09)	14	3.00 (3.55)	-0.856 (-2.167 to 0.456)

DOI: 10.3310/CWKF1987